Author's response to reviews

Title: Can work ability explain the social gradient in sickness absence: a cross-sectional study of a general working population in Sweden

Authors:

Jesper Löve (jesper.love@socmed.gu.se)
Kristina Holmgren (kristina.holmgren@neuro.gu.se)
Kjell Torén (kjell.toren@amm.gu.se)
Gunnel Hensing (gunnel.hensing@socmed.gu.se)

Version: 2 Date: 4 October 2011

Author's response to reviews:

Dear Editor,

In this revised version of the manuscript the abstract is now inserted. A sentence about ethical approval of the study is also inserted in the Methods section.

Kind regards,
Jesper Löve